Author's response to reviews

Title: Quantitative assessment of microbicide-induced injury in the ovine vaginal epithelium using confocal microendoscopy

Authors:

Gracie Vargas (grvargas@utmb.edu)
Igor Patrikeev (igpatrik@utmb.edu)
Jingna Wei (jiwei@utmb.edu)
Brent Bell (bbell@utmb.edu)
Kathleen L Vincent (klvincen@utmb.edu)
Nigel Bourne (nibourne@utmb.edu)
Massoud Motamedi (mmotamed@utmb.edu)

Version: 2 Date: 1 August 2011

Author's response to reviews:

As requested, we are submitting a revised abstract with the addition of a "Background" section and making sure the overall abstract word length not exceeding 350 words. We apologize for this oversight.